Author's response to reviews

Title: The Temporal Reliability of Serum Estrogens, Progesterone, Gonadotropins, SHBG and Urinary Estrogen and Progesterone Metabolites in Premenopausal Women

Authors:

Andrew E Williams (awilliam@crch.hawaii.edu)
Dr Gertraud Maskarinec (gertraud@crch.hawaii.edu)
Adrian A Franke (adrian@crch.hawaii.edu)
Frank Z Stanczyk (Fstanczyk@socal.rr.com)

Version: 4 Date: 6 Dec 2002

Equal contributorship - I clarified the sentence on the title page of the manuscript indicating which authors are equal contributors.
Insert figure tabs - I removed the "tabs" indicating where the figures should be inserted.